Author’s response to reviews

Title: Pulmonary Arterial Hypertension in a Patient Treated with Dasatinib

Authors:
Andris Skride (andris.skride@gmail.com)
Matiss Sablinskis (matiss.sablinskis@gmail.com)
Kristaps Sablinskis (kristaps.sablinskis@gmail.com)
Krista Lesina (krista.lesina@gmail.com)
Aivars Lejnieks (lejnieks@latnet.lv)
Sandra Lejniece (drsandra.lejniece@gmail.com)

Version: 1 Date: 19 Sep 2017

Author’s response to reviews:
1) The patients ethnicity has now been provided
2) The patient did not experience any relevant symptoms prior to April 2015
3) We did not add vital signs, because the patient was hospitalized multiple times, and there were no relevant findings to this case.
4) Relevant information on comorbidities and concomitant therapy has been added
5) Light's criteria was indeed used to investigate the pleural effusion fluid
6) Biochemistry data of the pleural fluid has been added
7) Table headings have been corrected accordingly